Author's response to reviews

Title: Prognostic implications of a carefully performed neurological assessment in patients with a first event suggestive of multiple sclerosis

Authors:

Jessica M Nielsen (jm.nielsen@vumc.nl)
Christoph Pohl (Christoph.Pohl@SCHERING.de)
Chris H Polman (CH.Polman@vumc.nl)
Frederik Barkhof (f.barkhof@vumc.nl)
Marc S Freedman (MFREEDMAN@ottawahospital.on.ca)
Gilles Edan (gilles.edan@chu-rennes.fr)
David Miller (d.miller@ion.ucl.ac.uk)
Lars Bauer (Lars.Bauer@SCHERING.DE)
Rupert Sandbrink (Rupert.Sandbrink@SCHERING.DE)
Ludwig Kappos (lkappos@uhbs.ch)
Bernard MJ Uitdehaag (bmj.uitdehaag@vumc.nl)

Version: 2 Date: 22 September 2008

Author's response to reviews:

Dear Sabina Alam,

Please find resubmitted the paper: prognostic implication of a neurological assessment in clinically isolated syndrome patients adapted to your suggestions.

- The title has been changed and now included multiple sclerosis
- A competing interest section has been included at the end of the paper
- An author’s contribution section has been added.
- The trial registration number has been filled.

Sorry for the delay and for not being complete,

Thank you for reviewing the manuscript,

Best wished,

Jessica Nielsen